

This is a repository copy of Measuring disability in multiple sclerosis: the WHODAS 2.0.

White Rose Research Online URL for this paper: <u>https://eprints.whiterose.ac.uk/203840/</u>

Version: Published Version

Article:

Young, C.A. orcid.org/0000-0003-1745-7720, Rog, D.J., Sharrack, B. et al. (6 more authors) (2023) Measuring disability in multiple sclerosis: the WHODAS 2.0. Quality of Life Research, 32 (11). pp. 3235-3246. ISSN 0962-9343

https://doi.org/10.1007/s11136-023-03470-6

Reuse

This article is distributed under the terms of the Creative Commons Attribution (CC BY) licence. This licence allows you to distribute, remix, tweak, and build upon the work, even commercially, as long as you credit the authors for the original work. More information and the full terms of the licence here: https://creativecommons.org/licenses/

Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.



eprints@whiterose.ac.uk https://eprints.whiterose.ac.uk/

Measuring disability in multiple sclerosis: the WHODAS 2.0

Carolyn A. Young^{1,2} • David J. Rog³ • Basil Sharrack⁴ • Cris Constantinescu⁵ • Seema Kalra⁶ • Tim Harrower⁷ • Dawn Langdon⁸ • Alan Tennant⁹ • Roger J. Mills^{1,2} on behalf of the Trajectories of Outcome in Neurological Conditions (MS) Study Group

Accepted: 21 June 2023 / Published online: 17 August 2023 © The Author(s) 2023

Abstract

Introduction Reliable measurement of disability in multiple sclerosis (MS) using a comprehensive, patient self-reported scale, such as the World Health Organization Disability Assessment Schedule (WHODAS) 2.0, would be of clinical and research benefit.

Methods In the Trajectories of Outcome in Neurological Conditions-MS study, WHODAS 2.0 (WHODAS-36 items for working, WHODAS-32 items if not working, WHODAS-12 items short-form) was examined using Rasch analysis in 5809 people with MS.

Results The 36- and 32-item parallel forms, and the cognitive and physical domains, showed reliability consistent with individual or group use. The 12-item short-form is valid for group use only. Interval level measurement for parametric statistics can be derived from all three scales which showed medium to strong effect sizes for discrimination across characteristics such as age, subtype, and disease duration. Smallest detectable difference for each scale was <6 on the standardised metric of 0–100 so <6% of the total range. There was no substantial differential item functioning (DIF) by age, gender, education, working full/part-time, or disease duration; the finding of no DIF for time or sample supports the use of WHODAS 2.0 for longitudinal studies, with the 36- and 32-item versions and the physical and cognitive domains valid for individual patient follow-up.

Conclusions Disability in MS can be comprehensively measured at interval level by the WHODAS 2.0, and validly monitored over time. Routine use of this self-reported measure in clinical and research practice would give valuable information on the trajectories of disability of individuals and groups.

Keywords Multiple sclerosis · Disability · Assessment · Rasch · Patient reported outcome measure · Trajectories of Outcome in Neurological Conditions (TONiC) · World Health Organization Disability Assessment Schedule 2.0

Plain English summary

Patient care and research in multiple sclerosis (MS) would be greatly improved if it was possible to comprehensively measure disability by patient self-report. The World Health

Carolyn A. Young Carolyn.young11@nhs.net

- ¹ Walton Centre NHS Foundation Trust, Lower Lane, Fazakerley, Liverpool L9 7LJ, UK
- ² Institute of Systems, Molecular and Integrative Biology, University of Liverpool, Liverpool, UK
- ³ Northern Care Alliance NHS Trust, Salford, UK
- ⁴ Academic Department of Neurology, University of Sheffield, Sheffield, UK

Organisation developed the self-reported Disability Assessment Schedule 2.0 (WHODAS), measuring cognition, mobility, self-care, getting along, life activities, and participation. In this study of 5809 people with MS taking part in the TONiC study, we showed that the WHODAS can be

- ⁵ University of Nottingham, Nottingham, UK
- ⁶ University Hospital of North Midlands NHS Trust, Stoke-on-Trent, UK
- ⁷ University of Exeter Medical School, Exeter, UK
- ⁸ Royal Holloway, University of London, Egham, Surrey, UK
- ⁹ Leeds Institute of Rheumatic and Musculoskeletal Medicine, University of Leeds, Leeds, UK



reliably used in MS, importantly, results can be easily converted to a standardised format which allows change to be measured. We checked two versions of the WHODAS, one for those who are not working (32 questions) and another for those in work/education (adds 4 questions). We proved these can be used interchangeably, so disability could be followed for individuals or groups over years as they move between education, work and not working. The short-form (12 questions) is accurate only for groups, not individuals. It would be simple to add regular assessment with the WHO-DAS to routine follow-up, providing valuable information to guide care.

Introduction

The COVID19 pandemic has accelerated the move to increased remote assessment and monitoring of people with multiple sclerosis (pwMS). This has strengthened interest in identifying reliable and simple methods of measuring disability which do not require pwMS to attend in person, or clinicians to attempt disability assessment by phone or video.

Can measures traditionally assessed through clinical examination be adapted? The Expanded Disability Status Scale (EDSS) has been investigated for patient self-report in varied formats [1] but has psychometric weaknesses (see review [2]). Furthermore, a qualitative study with pwMS found that cognition and participation, topics not well measured by the EDSS, were important components of MS disability [3]. A more recent qualitative study of patients' perceptions of outcome measures found that they should be comprehensive and measure neurological symptoms, cognitive impairments, mental health and well-being, self-care activities and social challenges, in order to adequately capture and support the needs of pwMS [4]. This requires either a range of patient-reported outcome measures (PROMs) or PROMs with sufficient breadth of coverage.

The World Health Organization (WHO) developed the WHO Disability Assessment Schedule WHODAS 2.0 as a comprehensive measure of disability, suitable for clinical practice or measurement at the population level [5]. It has been widely adopted; a systematic review on the use of the WHODAS 2.0 identified 810 studies [6]. All versions of the WHODAS 2.0 cover six domains: cognition, mobility, selfcare, getting along, life activities, and participation. In the 36-item version (hereafter termed WHODAS-36), life activities encompass both household and work responsibilities. For use with people who are no longer working, four items related to work have been removed from this 'life activities' domain to create the 32-item WHODAS 2.0 (hereafter termed WHODAS-32). In order to create a more concise measure, two items for each of the six domains have been

used to create the 12-item WHODAS 2.0 (hereafter termed WHODAS-12) [7].

There has been relatively little work exploring the structural validity of the WHODAS 2.0 in pwMS. One exploratory factor analysis of the WHODAS 2.0 in the Persian version for MS patients identified seven factors but the Confirmatory Factor Analysis did not comply with published requirements for acceptable fit [8, 9]. Another study looked at the earlier WHODAS II scale using Rasch analysis in 136 pwMS and found it a reliable and valid instrument for the assessment of patient-reported disability in MS [10].

The use of generic as opposed to disease-specific measures has been long discussed [11]. One advantage of generic measures such as the WHODAS 2.0 is that comparisons with non-MS populations are possible. The Nurses' Health Study found that women with MS had a consistently lower physical ability than unaffected women [12]. Physical functioning gradually decreased with increasing age in both groups, but pwMS declined 3–4 times faster in midlife, while decline was similar in old age. This would indicate a relatively rapid decline in physical functioning during the key period when engaged with work. As an example, the physical function score of 45-year-old women with MS was comparable to that of 75-year-old unaffected women [12].

The current study examines the structure of the WHO-DAS 2.0 in its various formats in a calibration sample of 1050 MS patients recruited into the longitudinal Trajectories of Outcome in Neurological Conditions (TONiC) study, applying the Rasch measurement model [13]. It then describes the various WHODAS 2.0 scores across key demographic and clinical factors in the full sample of 5809 pwMS. Trajectories of disability in both an inception cohort and the full sample are explored. The aims are to investigate the capacity of the patient self-reported WHODAS 2.0 to measure the disability in pwMS at individual and group level, ideally enabling generation of interval level data which can be validly analysed by parametric statistics, with adequate effect sizes across both demographic and clinical groups.

Methods

Samples

Participants were recruited into the Trajectories of Outcome in Neurological Conditions-MS (TONiC-MS) study from hospitals and community teams in 33 collaborating sites across the UK (https://tonic.thewaltoncentre.nhs.uk/). Eligibility criteria included adults (aged over 18 years, and with no upper age limit) with physician-verified MS (by McDonald criteria [14]) of any disease subtype and level of disability, providing they could give informed consent and complete questionnaire packs (with the help of a scribe if necessary).

Data on disease subtype at time of study entry were provided by clinicians involved in the patients' care and classified as relapsing–remitting (RRMS), primary progressive (PPMS), and secondary progressive (SPMS). Duration since diagnosis and Expanded Disability Status Scale (EDSS) band was recorded from the medical records. Informed consent was obtained from all participants prior to enrolment.

Participants completed a baseline questionnaire pack containing a number of PROMs (for list, see study protocol schematic in Supplementary File 1). Follow-up questionnaire packs were sent at intervals of at least 9 months. In total, the baseline and up to four follow-ups were included in the current data cut.

For analytical purposes, additional samples were drawn from the overall sample. The Rasch analysis used a calibration sample of 1050 cases, where 350 cases were randomly selected from each of first three time points in such a way that no individual appears in the total sample more than once [15]. The calibration sample was drawn from participants who had completed all 36 items of the WHODAS 2.0, so omitted those who had missing values on the four work items (except where they were imputed, see below). In addition, the data were further randomised into 'training' and 'validation' samples for cross-validation purposes. An inception cohort was also drawn from the full sample, being those pwMS with a duration 2 years or less years.

Ethical approval was granted from research committees (reference 11/NW/0743).

Outcome measures

WHODAS 2.0

The scoring instructions for the WHODAS 2.0 define three specific scores: (1) items within a domain are summated to give the simple domain score; (2) domains are summated to give the simple total score; and (3) the total score is standardised to 0-100 [16]. The simple score range of the WHODAS-36 is 0-144, 0-128 for the WHODAS-32, and 0-48 for the WHODAS-12, where higher scores measure worse disability.

For the WHODAS-36, where respondents chose to skip the four items related to work and also reported that they were medically retired, then the item responses for the set were imputed to the response 'extreme or cannot do'. Otherwise, items affected by the skip were flagged as missing. The same imputation was made to the one work item in the WHODAS-12. Only the 36 items were presented to the pwMS, and analysis then proceeded on the relevant set of items for each version.

EQ-5D-5L

The EQ-5D-5L utility value derived from 5 items scored 1-5; the range is from -0.285 to 1, where higher scores indicate better health states [17, 18].

Neurological Fatigue Index-MS (NFI-MS)

The 10-item summary scale scored 0–30, where higher scores represent greater MS fatigue [19].

Rasch analysis

Briefly, the data from each (sub)scale in each version of the WHODAS 2.0 were fit to the Rasch measurement model, to determine the internal construct validity, reliability, and invariance to key contextual factors [13]. These included age, gender, subtype of MS, duration, and time (baseline and first two follow-ups). Training and validation samples were included to further support invariance for the cross-validation analysis.

Full details of the Rasch methodology applied can be found in Supplementary File 2. Briefly, a hierarchical analysis strategy was employed starting with the summation of individual items (Level 1), through subscale scores or clusters of locally dependent items (Level 4), to parallel forms (Level 6) (Supplementary File 2: Table S1). The requirements of the Rasch model, elucidated in Supplementary File 2, must be met for any level of solution. A conceptual-based component approach can also be used where domains show local dependency, and these can be grouped into some overarching concept such as physical or social. Furthermore, at any level of analysis with local item dependency, a bi-factor approach can be utilised when dependency has been accommodated. This approach is based on the concept of essential unidimensionality and derives its estimate from what can be considered as the first common factor. The proportion of variance retained in that solution is reported.

Trajectory analysis

To illustrate the use of the WHODAS 2.0 as a measure of overall disability, the total standardised score for the WHODAS-36 was examined over time of follow-up for the inception cohort and WHODAS-32 for the full sample, the latter adjusted for duration at baseline. Full details of the methodology are given in Supplementary File 2. Briefly, the time metric was the median month since the baseline questionnaire at each follow-up. Disability was assessed at baseline and up to four further follow-ups and modelled with a censored normal distribution.

Additional attributes

The standard error of measurement (SEM) was calculated as $SD^*\sqrt{(1 - reliability)}$ and the smallest detectable difference (SDD) as $\pm 1.96^*\sqrt{2^*SEM}$. The latter is the smallest statistically significant change in measurement results. The magnitude of effect sizes followed Cohen's recommendations [20]. For the Rasch analysis, the effect size for substantial differential item functioning followed the simulation from Rouquette [21].

Participants were categorised as early onset if their date of diagnosis was before age 18 years and late onset if diagnosis was made after age 50 [22, 23]. The Index of Multiple Deprivation score gives an overall score for the relative level of multiple deprivation by area of residence of the participant [24].

Finally, in the first follow-up questionnaire, patients were asked to state whether or not their health had become worse, stayed the same, or improved. The same applied to their disability. Thus an anchor-based responsiveness was calculated, based upon the different metric estimate of those reporting worse or better. An effect size of that difference was given. Test–retest reliability was also calculated based upon those indicating no change in their disability state.

Results

The samples

The characteristics of the full and inception cohorts are summarised in Table 1. The mean EQ-5D-5L utility value of those in the inception cohort (duration since MS diagnosis 2 years or less) was significantly better than those in the full cohort (0.762, SD 0.217, compared to 0.680, SD 0.253; t = -9.97 (*df* 5795); $p \le 0.001$). Over two-fifths (43.8%) of those with RRMS were on disease-modifying therapy (DMT).

The mean EQ-5D-5L utility value varied considerably by disease subtype, ranging from 0.763 (SD 0.206) in RRMS to 0.511 (SD 0.261) in SPMS. There was also a significant difference in EDSS level by disease subtype; for example, EDSS level 0–4 ranged from 8.8% in SPMS, to 71.4% in RRMS (χ^2 2.1e+03(9): p < 0.001).

For the calibration sample, baseline age, duration, and health utility value were not significantly different to the remainder of the full sample (*t*-test p > 0.05), neither were gender, type of onset nor EDSS levels ($\chi^2 p > 0.05$).

Table 1 Demographic and clinical characteristics of the different samples

Sample/characteristics	Full	Inception
N	5809	813
Mean age (SD)	50.1 (12.0)	43.2 (12.1)
% Female	73.8	72.4
Mean duration since diagnosis in years (SD)	11.1 (9.8)	0.6 (0.5)
% Subtype		
Relapsing remitting	65.9	81.8
Secondary progressive	23.0	4.8
Primary progressive	11.2	13.4
% EDSS band		
0–4	51.0	74.5
4.5-6.5	37.5	22.6
7–7.5	6.7	2.5
8–9.5	4.7	0.4
Mean EQ-5D-5L utility value (SD)	0.680 (0.253)	0.762 (0.217)

The Rasch analysis of WHODAS-36, WHODAS-32, and WHODAS-12

The total scores of all three versions showed adequate fit to the Rasch model. Reliability (alpha) was consistent with individual clinical use except for the WHODAS-12, which only supported group use. A summary of the strategy of fit of the data in the calibration sample from each version of the WHODAS 2.0 and its various domains is given in Table 2. Briefly, a level 1 solution is at the item level, level 4 is two clusters of domains, or two clusters of LD items, level 5 is by taking alternative items, and level 6 requires item deletion (for full details, see Supplementary File 2.1). Data were also analysed after grouping domains into two components: 'cognitive-social' incorporating the three domains of cognition, getting along and participation, and 'physical', incorporating the remaining three domains of mobility, self-care, and life activities.

Applying the different levels of strategy for fit it was possible to fit the data to the Rasch model. The bi-factor equivalent solutions for the total scores of the WHODAS-36 and WHODAS-32 only discarded 8% of the variance to provide a unidimensional latent estimate, whereas the WHODAS-12 discarded more (average 13% across the two training and validation samples).

Detailed analyses for all results are provided in Supplementary File 3. Transformation tables for the different versions are available in Supplementary File 4, utilising the complex standardisation approach, giving each version a score of 0–100, to allow comparison across versions.

Figure 1 shows the distribution of the three versions using the standardised score using a Kernel density estimate.

Table 2Solution strategy of fitto the Rasch model of domains,components, and total scores ofall versions of the WHODAS2.0

	Simple item summation		Bi-factor equivalent solution		
	With reliabil- ity≥0.85	With reliability <0.85	With reliability ≥0.85	With reliability < 0.85	
WHODAS-36					
Domains					
Cognition	Items				
Mobility	Items				
Self-care				Super Items	
Getting along		Item deletion			
Life activities				Testlet	
Participation			Testlet		
Components					
Cognitive-social				Testlet	
Physical		Items			
Total			Testlet		
WHODAS-32					
Total			Testlet		
WHODAS-12					
Components					
Cognitive-social				Testlet	
Physical	Items				
Total				Testlet	

WHODAS World Health Organization Disability Assessment Schedule; *Items* Fit at the item level; *Testlet* Fit with testlets; *Super Items* Fit with alternative items; *Item deletion* remove one or more items





^aWHO36/32/12: World Health Organization Disability Assessment Schedule 2.0 36/32/12 items

While the WHODAS-36 and WHODAS-32 are equivalent, this is not the case for the WHODAS-12, which has a much lower mean score, and hence, its distribution is left-shifted from the other versions. Where the WHODAS-12 scores zero if the subject truly had no disability, they should also score 0 on the WHODAS-32; however, this was not the case and for those subjects with WHODAS-32 score > 10, items not in the WHODAS-12 such as 'Remembering to do important tasks' or 'How much time did you spend on your health

condition, or its consequences' were gaining points on the WHODAS-32.

Discrimination of WHODAS 2.0 versions across key demographic and clinical characteristics

With the exception of gender, all three parallel forms of the WHODAS 2.0 showed medium to strong effect sizes for discrimination across key characteristics (Table 3). The
 Table 3
 Descriptive analysis

 of the different versions of the
 WHODAS 2.0 (standardised

 metric). Full sample at baseline
 Standardised

Attribute	WHODAS-36	WHODAS-32	WHODAS-12	Ν
Age group (years)				
<44 [A]	37.3	36.5	27.8	1634
44–51	40.6	40.0	32.3	1466
52–59	43.1	42.6	35.5	1404
60+[D]	43.8	43.7	36.7	1299
Effect size A:D	0.56	0.69	0.66	
Gender				
Male	41.8	41.2	33.3	1521
Female	40.7	40.2	32.6	4281
Effect size	0.10	0.14	0.06	
MS subtype				
Primary progressive	44.5	44.6	37.6	649
Relapsing remitting	38.4	37.5	29.3	3823
Secondary progressive	46.7	47.1	40.6	1331
Effect size RR:SP	0.85	0.94	0.96	
EDSS				
0–4.0 [A]	35.8	34.6	25.8	2963
4.6-6.5	45.8	45.7	38.9	2175
7.0–7.5	48.9	49.7	43.5	390
8.0–9.5 [D]	49.2	50.5	44.0	275
Effect size A:D	1.41	1.66	1.62	
Duration (years)				
0–2 [A]	37.9	37.2	28.3	1201
3–8	39.4	38.6	30.7	1641
9–16	42.1	41.6	34.3	1543
17+[D]	44.3	44.1	37.3	1418
Effect size A:D	0.57	0.59	0.65	
Total (mean)	41.0	40.5	32.8	5803

discrimination was particularly strong across the EDSS, where the WHODAS-32 was marginally more discriminating than other versions, albeit all versions had a 'strong' effect size. The difference of means at baseline between WHODAS-36 and WHODAS-32 gave an effect size of 0.04, indicating that they are equivalent.

Standard error of measurement (SEM) and the smallest detectable difference (SDD)

The standard error of measurement (SEM) and the smallest detectable difference (SDD) of the WHODAS-36 was 1.971 and 5.466, respectively, on the standardised metric. For the WHODAS-32, the SEM was 2.068 and SDD 5.733, and for the WHODAS-12, the SEM was 2.360 and SDD 6.543. As

these are all on the standardised metric, then the percent of the full operational scale required to get above error is the same as the SDD. In contrast, the % SDD for the EQ-5D-5L was 15.533 based upon a reported 0.92 ICC reliability [25].

Responsiveness to change across baseline and first follow-up questionnaire

Table 4 shows the responsiveness of the different versions of the scale, and the effect size of the magnitude of difference reported between those reporting worse and better. For health, 27.4% reported a worse state while 11.6% reported better. For disability, 40.9% reported a worse state, and 5.7% reported better. All versions of the scale across both anchors showed a large effect size.

 Table 4 Effect size of metric differences between those reporting health or disability as worse or better

	Worse	Same	Better	Effect size
WHODAS-36				
Health	1.535	0.298	-1.810	0.671
Disability	1.385	0.023	-3.141	0.895
WHODAS-32				
Health	1.797	0.230	-1.492	0.656
Disability	1.689	-0.074	-3.144	1.030
WHODAS-12				
Health	2.332	0.990	-1.482	0.531
Disability	2.053	0.756	-2.759	0.645

Test retest reliability based upon those indicating no change in disability was 0.81, 0.82 and 0.83 for the WHO-DAS-36 -32 and -12, respectively.

Inception cohort trajectory analysis using WHODAS-36

Given the estimates derived from the Rasch analysis, an exploration was made as to how, if at all, the average levels of disability masked underlying groups with different trends over time. Two distinct WHODAS-36 trajectories were observed for those in the inception cohort (Fig. 2). The majority (88.2%) displayed a moderate level of disability with a gradual increase in the level of disability, but just over one in ten (11.8%) displayed a low level of disability, which remained low over time. Almost all of these low disability group (93.2%) were RRMS and were significantly younger than those with higher disability (39.0 vs 43.8 years) (t - 3.52 (df 811); $p \le 0.001$). There was no significant difference in gender, nor use of DMT between groups.

Full sample trajectory analysis using WHODAS-32

Three trajectories were identified which satisfied the conditions given in Supplementary File 2 (Fig. 3). All three trajectories showed strong discrimination across all demographic, clinical, and other factors (Table 5). Of note, Groups 1 and 2, which showed quite different levels of disability over time, had equivalent durations. On the other hand, Group 3 members were much older, with a longer duration, and just 29.7% were in work, compared with 79.2% of those in Group 1.

A multinomial logistic regression explored which demographic, clinical, and other contextual factors may differentiate Groups 2 and 3 from Group 1 (Tables 6, 7). In summary, Groups 2 and 3 are older than those in Group 1, and more likely to be religious, more likely to have comorbidities, and be past and current smokers. Group 3 members are much less likely to be RRMS than Group 1 and to live in areas with higher deprivation. Of note, the relative risk of stage of onset (early or late) does not significantly vary between Groups 2 and 3, and Group 1. The Cragg–Uhler (Nagel-kerke) pseudo- R^2 was 0.27, indicating an adequate difference to the base model.

Discussion

The current study demonstrates that the total scores of the various versions of the WHODAS 2.0 can be used to measure disability in MS, through fit to the Rasch measurement model utilising a bi-factor equivalent approach. Thus, the WHODAS 2.0 provides a simple patient-reported outcome measure of disability, suitable for use by pwMS. Use of the transformation tables gives each version a score of 0-100, to allow comparison across versions; these interval level scores can validly be used for parametric analyses including change scores. The WHODAS-36 and WHODAS-32 can be seen as equivalent, with the latter used for people who are not working. If an individual being followed over the disease course moves between work/education and non-working status, their disability can be tracked seamlessly, using the WHODAS-36 during work/education and the WHODAS-32 when not working. Similarly, in a mixed sample of those in work or not, the appropriate transformation from the nomogram in Supplementary File 3 could be used for each version, obtaining the standardised score for all. This capacity to track disability over time, and across individuals and populations, allows analysis of changes linked to clinical or public health interventions. The total scores in both the WHODAS-36 and WHODAS-32 were robust enough for individual use. In contrast, the score on the WHODAS-12 cannot be considered equivalent to the larger versions; its reliability permits group use but is insufficient for individual use, and it may under-estimate some aspects of disability due to its abbreviated item set. All three versions of the scale showed good responsiveness to change.

The WHO conceptualises that every person is on a continuous spectrum of disability, or functioning, ranging from no disability/full functioning. Thus, WHO would not categorise individuals as disabled or not disabled, but place them on this continuous spectrum of functioning which can be measured using the WHODAS 2.0. The level of disability, or functioning, can be measured irrespective of the aetiology. For example, a person with MS and arthritis can have their mobility measured, and compared against those with MS alone or arthritis alone. The assessor is not asked to attempt to qualify how much mobility impairment is due to MS and how much to arthritis, or assume that comorbidities have no impact and disability relates to MS alone.



^aWHODAS-36: World Health Organization Disability Assessment Schedule 2.0-36 items



Fig. 3 WHODAS-32 trajectories: full sample

^aWHODAS-32: World Health Organization Disability Assessment Schedule 2.0-32 items

When disease-specific disability scales are employed, imprecision may occur when people are followed longitudinally if all disability is attributed to the disease and the effects of ageing and comorbidities disregarded. For example, if pwMS are followed over their life course, EDSS will be affected by natural changes of ageing, such as bilateral absent ankle reflexes in 8% of all adults aged 51-60 years and 30% of those aged 61-70, which would affect scoring of the EDSS pyramidal function [26]. When testing healthy subjects with no comorbidities or risk factors for sensory abnormality, 29% of over 65-years-olds have absent vibration sense at most distal joint compared to 1% of 18-64-yearolds [27]. This finding, occurring in over a quarter of healthy over 65 year olds, results in a score of 3 on EDSS sensory function. In a study measuring EDSS in 106 people, 55 years and older, with and without MS, median EDSS scores were 6.0 in people with MS and 3.0 in people without MS [28].

The WHO developed the WHODAS 2.0 to measure the person's performance in a range of domains—cognition, mobility, self-care, getting along, life activities, and participation—which cover the breadth of human functioning. These domains of the WHODAS 2.0 mostly showed adequate fit although life activities had poor fit and reliability. However, the level of reliability of domains in the

WHODAS-12 required a component approach to model fit, combining cognition, getting along and participation as one component termed 'cognitive-social', and mobility, selfcare, and life activities as the other component termed 'physical'. This offered some improvement in fit at the component level, and a better solution when the two components were used together to examine the total score. Thus, the WHO-DAS-12 is suitable for group use but not for tracking the disability, or functioning, of an individual over time.

The mobility and cognition domains in both the WHO-DAS-36 and WHODAS-32 were robust enough for individual use. Therefore, the cognition domain may have potential for simple, rapid monitoring of cognition-related disability in MS. A recent systematic review of the social consequences of MS disability found higher work disability in relation to higher physical disability and lower cognitive function [29]. Another study using latent class modelling of longitudinal cognitive data identified patients with worsening reaction times and increased risk of disability progression [30]. The authors argued that monitoring of cognition in clinical practice may enable detection of cognitive change trajectories and people with RRMS at risk of disability progression.
 Table 5
 Baseline characteristics
 of three trajectory groups: full

sample using WHODAS-32

Table 6 Contrast between

trajectory Group 2 and Group 1

	Trajectory Group			Significance		
	1	2	3	Total	Р	Ν
WHODAS-32 (SD)	10.3 (8.9)	31.5 (6.0)	46.8 (5.4)	40.5 (11.9)	< 0.001	5799
Age in years (SD)	43.3 (11.2)	46.8 (11.8)	52.1 (11.6)	50.2 (12.0)	< 0.001	5809
Duration in years (SD)	7.6 (7.6)	7.5 (7.7)	12.7 (10.3)	11.1 (9.8)	< 0.001	5809
% Female	77.4	75.2	72.9	73.8	0.054	5808
MS subtype						
PP (%)	2.8	7.2	13.5	11.1		
RR (%)	94.7	87.0	55.3	65.9	< 0.001	5809
SP (%)	2.5	5.8	31.3	23.0		
EDSS (%)						
0–4.0	94.2	85.5	34.2	51.0		
4.5-6.5	5.1	13.8	49.4	37.5		
7.0–7.5	0.5	0.4	9.6	6.7	< 0.001	5809
8.0–9.5	0.2	0.3	6.8	4.7		
Health utility (mean)	0.949 (0.101)	0.871 (0.100)	0.583 (0.244)	0.680 (0.253)	< 0.001	5693
Fatigue NFI-MS (SD)	7.6 (5.6)	14.0 (5.1)	21.0 (5.2)	18.4 (6.7)	< 0.001	5545
In work (%)	79.2	66.5	29.7	42.2	< 0.001	5809
Ν	394	1446	3969	5809		

Subtype PP primary progressive; RR relapsing remitting; SP secondary progressive; EDSS Expanded Disability Status Scale; NFI-MS Neurological Fatigue Index-MS

All group comparisons are significant ($P \le 0.001$) with the exception of % female

Aspect	Relative risk	SE	Р	Confidence Intervals	
				Lower	Upper
Group 2 relative to group 1					
Age	1.029	0.009	0.002	1.011	1.047
Male	1.089	0.179	0.605	0.789	1.502
Married	0.864	0.129	0.329	0.644	1.159
Religious/spiritual	1.416	0.225	0.028	1.038	1.933
Onset stage—early	2.070	1.238	0.224	0.641	6.683
Onset stage—late	1.009	0.296	0.974	0.572	1.781
Relapsing remitting	0.547	0.205	0.108	0.263	1.141
Secondary progressive	1.176	0.613	0.756	0.423	3.266
Duration	0.967	0.011	0.003	0.946	0.989
Comorbidity	1.715	0.238	< 0.001	1.308	2.251
Smoke—past	1.639	0.251	0.002	1.205	2.204
Smoke—current	1.930	0.512	0.013	1.147	1.781
Multiple deprivation	1.007	0.005	0.203	0.996	1.018

Multinomial logistic regression

Multiple Deprivation: Index of Multiple Deprivation 2011

The WHODAS 2.0 showed good effect sizes across different ages, MS subtypes, EDSS band, and duration of disease. Trajectory analysis among 5809 people with MS showed three trajectories, which varied in their baseline characteristics of age, disease duration, MS subtype, EDSS band, health utility, fatigue, employment status,

Table 7 Contrast between trajectory group 3 and group 1

		C	Quality of Life Rese	earch (2023) 32	2:3235–3246
Aspect	Relative risk	SE	Р	Confidence Intervals	
				Lower	Upper
Group 3 relative to Group 1					
Age	1.029	0.009	0.001	1.011	1.047
Male	1.167	0.186	0.334	0.853	1.596
Married	0.792	0.115	0.108	0.595	1.053
Religious/spiritual	1.680	0.259	0.001	1.242	2.274
Onset stage—early	0.989	0.600	0.985	0.301	3.248
Onset stage—late	0.742	0.211	0.297	0.425	1.298
Relapsing remitting	0.162	0.586	< 0.001	0.080	0.329
Secondary progressive	2.035	1.017	0.155	0.765	5.419
Duration	1.015	0.011	0.150	0.995	1.036
Comorbidity	3.041	0.412	< 0.001	2.332	3.966
Smoke—past	1.898	0.285	< 0.001	1.414	2.547

0.942

0.005

factors such as age by gender.

Multinomial Logistic Regression

Smoke-current

Multiple deprivation

Multiple Deprivation: Index of Multiple Deprivation 2011

3.689

1.018

comorbidity, smoking history, religious or spiritual belief, and level of socio-economic deprivation. This would indicate that the 'average' trajectory of disability masks potentially important factors relevant to clinical management. For example, in the full sample while Groups 1 and 2 had similar durations, Group 1 showed a much lower level of disability, which was retained over four years. Does this indicate a possible 'benign' trajectory or different prodromal experience? In the inception cohort, a similar group with low disability was identified. What consideration should be given to the longer-term management of those who retain a low level of disability over time? A key implication of these findings is that any intervention based upon 'average' changes in disability may mask important differences in clinical path or influences of socio-economic factors, such as access to health care, which should be explored further [30].

Modelling using various registries has described mild, moderate, and severe disability trajectories in both PPMS and SPMS [31, 32]. A Swedish study examined trajectories of mean Sickness Absence and Disability Pension days per year (SA/DP) and found higher levels of SA/DP among pwMS of working age compared to the reference population [33]. Three trajectory groups of SA/DP were identified in pwMS: persistently low (55.2%), moderate increasing (31.9%), and high increasing (12.8%) [33].

There are a number of limitations to the study. As with all longitudinal studies, there is the problem of attrition, with the baseline and four further follow-ups requiring data collection 5 times over many years. The attrition appears worse for two reasons. The first is the average 22-month time between baseline and first follow-up. This was due to

the administrative tasks of obtaining all the necessary ethical approvals to activate the follow-up stage and in the early part of the study resulted in a significant loss of participants. The second is that the current analysis is based upon a data cut in October 2019, and TONiC is an ongoing study. Therefore, what we have termed in a purist sense 'attrition' documents those individuals who had not provided followup data at that time point. However, the trajectory analysis accounted for differences in attrition across groups, and all other descriptive analyses were based upon the baseline data. The analysis of DIF also uses just single factors such as age or gender. No attempt was made to provide more complex

< 0.001

0.001

2.237

1.008

6.084

1.028

The strengths of the study include the sample size; the calibration sample of 1050 people for the Rasch analysis, which delivered invariance for time and sample, together with interval scale data for analysis. The Rasch model itself is a strength, as it complies with fundamental measurement requirements, which include non-intersecting Item Characteristic Curves, and homogeneity, both of which are not available from other parametric IRT models. As a consequence, it provides traceability, a standard unit within its frame of reference, in this case pwMS, so allowing reliable and valid comparisons when measuring person ability [34]. Furthermore, the use of the WHODAS 2.0 itself, which has additional advantages in that it is built on the International Classification of Functioning, Disability, and Health (ICF) which is one of the recommended systems for e-health informatics [35–37]. Finally, this work provides transformation tables which allow users of the WHODAS 2.0 to transform their ordinal raw scores

to interval level estimates based on the complex scoring metric of 0-100.

The study also raises issues for further research. It has been proposed to refine the concept of the MS continuum to include a prodromal phase which will help inform the true "at risk" period when considering exposures that might cause MS [38]. MS registries may have a part to play in this, although they can be limited in the range of information collected [39]. While different registries and studies may collect different information on health status, test equating may help to overcome the use of different PROMs, equating them onto a common reference metric [40]. In addition, the linking of the genome to the lived experience of MS explored through studies like TONiC may uncover genetic risk factors associated with the condition and the influence on how the disease manifests over the life course.

Conclusions

These results show that the WHODAS 2.0 has considerable strengths for disability assessment and monitoring in pwMS. It can be collected by patient self-report and used at individual or group level to follow pwMS through life stages such as education, work or not working, without distortion from DIF and with a small SDD. Raw scores can rapidly be converted to interval level measurement, allowing use of parametric statistics such as means and change scores. It showed moderate to good effect sizes for key characteristics. The wide use of the WHODAS 2.0 in other diseases and healthy populations allows bench marking of disability and comparison to other populations. Incorporation of the WHODAS 2.0 into routine clinical follow-up and research would provide valuable information to guide care and contextualise research findings.

Supplementary Information The online version contains supplementary material available at https://doi.org/10.1007/s11136-023-03470-6.

Acknowledgements The authors thank the participants and their families for their invaluable contributions, the research and clinical staff for recruitment, and the TONiC team. The authors thank Multiple Sclerosis Society (Grant No. C009-16.1), Biogen Idec, Genzyme, Merck Serono, Novartis, Roche, Teva, and Neurological Disability Fund 4530, for financial support.

Funding The author(s) disclosed receipt of the following financial support for the research, authorship, and/or publication of this article: This work was supported by the Multiple Sclerosis Society (Grant No. C009-16.1), Biogen Idec, Genzyme, Merck Serono, Novartis, Roche, Teva, and Neurological Disability Fund 4530. The authors also thank the NIHR CLRN for research support.

Declarations

Conflict of interest The author(s) declared no potential conflict of interest with respect to the research, authorship, and/or publication of this article.

Data Access Data supporting this study are not openly available due to reasons of sensitivity and are available from the corresponding author upon reasonable request. Data are located in controlled access data storage at Walton Centre NHS Trust. Please contact wcft.tonic@nhs. net.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Collins, C. D., Ivry, B., Bowen, J. D., Cheng, E. M., Dobson, R., Goodin, D. S., et al. (2016). A comparative analysis of Patient-Reported Expanded Disability Status Scale tools. *Multiple Sclerosis*, 22(10), 1349–1358.
- Hobart, J., Freeman, J., & Thompson, A. (2000). Kurtzke scales revisited: the application of psychometric methods to clinical intuition. *Brain*, 123(Pt 5), 1027–1040.
- Coenen, M., Basedow-Rajwich, B., König, N., Kesselring, J., & Cieza, A. (2011). Functioning and disability in multiple sclerosis from the patient perspective. *Chronic Illness*, 7(4), 291–310.
- Westergaard, K., Skovgaard, L., Magyari, M., & Kristiansen, M. (2022). Patient perspectives on patient-reported outcomes in multiple sclerosis treatment trajectories: A qualitative study of why, what, and how? *Multiple Sclerosis and Related Disorders*, 58, 103475.
- Üstün, T. B., Chatterji, S., Kostanjsek, N., Rehm, J., Kennedy, C., Epping-Jordan, J., et al. (2010). Developing the World Health Organization Disability Assessment Schedule 2.0. Bulletin of the World Health Organization, 88(11), 815–823.
- Federici, S., Bracalenti, M., Meloni, F., & Luciano, J. V. (2017). World Health Organization disability assessment schedule 2.0: An international systematic review. *Disability and Rehabilitation*, 39(23), 2347–2380.
- Andrews, G., Kemp, A., Sunderland, M., Von Korff, M., & Ustun, T. B. (2009). Normative data for the 12 item WHO Disability Assessment Schedule 2.0. *PLoS ONE*, 4(12), e8343.
- Salehi, R., Negahban, H., Khiavi, F. F., Saboor, S., Majdinasab, N., & Shakhi, K. (2020). Validity and reliability of the World Health Organization Disability Assessment Schedule 2.0 36-Item Persian version for persons with multiple sclerosis. *Korean Journal of Family Medicine*, 41(3), 195–201.
- 9. Kline, R. (2011). *Principles and Practice of Structural Equation Modeling* (3rd ed.). Guilford Press.
- Magistrale, G., Pisani, V., Argento, O., Incerti, C. C., Bozzali, M., Cadavid, D., et al. (2015). Validation of the World Health Organization Disability Assessment Schedule II (WHODAS-II)

in patients with multiple sclerosis. *Multiple Sclerosis Journal*, 21(4), 448–456.

- Patrick, D. L., & Deyo, R. A. (1989). Generic and disease-specific measures in assessing health status and quality of life. *Medical Care*, 27(3 Suppl), S217–S232.
- Cortese, M., Bjornevik, K., Chitnis, T., Ascherio, A., & Munger, K. L. (2022). Aging with multiple sclerosis: A longitudinal study of physical function, mental health, and memory in two cohorts of US women. *Multiple Sclerosis Journal*, 28(1), 121–131.
- 13. Rasch, G. (1960). Probabilistic Models for Some Intelligence and Attainment Tests. University of Chicago Press.
- Polman, C. H., Reingold, S. C., Banwell, B., Clanet, M., Cohen, J. A., Filippi, M., et al. (2011). Diagnostic criteria for multiple sclerosis: 2010 revisions to the McDonald criteria. *Annals of Neurology*, 69(2), 292–302.
- Mallinson, T. (2011). Rasch analysis of repeated measures. *Rasch Measurement Transactions*, 25(1), 1317.
- Üstün, T., Kostanjsek, N., Chatterji, S., & Rehm, J. (2010). Measuring Health and Disability Manual for WHO Disability Assessment Schedule. World Health Organization.
- Herdman, M., Gudex, C., Lloyd, A., Janssen, M., Kind, P., Parkin, D., et al. (2011). Development and preliminary testing of the new five-level version of EQ-5D (EQ-5D-5L). *Quality of Life Research*, 20(10), 1727–1736.
- Devlin, N. J., Shah, K. K., Feng, Y., Mulhern, B., & van Hout, B. (2018). Valuing health-related quality of life: An EQ-5D-5L value set for England. *Health Economics*, 27(1), 7–22.
- Mills, R. J., Young, C. A., Pallant, J. F., & Tennant, A. (2010). Development of a patient reported outcome scale for fatigue in multiple sclerosis: the Neurological Fatigue Index (NFI-MS). *Health and Quality of Life Outcomes*, 8, 22.
- 20. Cohen, J. (1988). Statistical Power Analysis Hillsdale. Erlbaum.
- Rouquette, A., Hardouin, J. B., Vanhaesebrouck, A., Sébille, V., & Coste, J. (2019). Differential Item Functioning (DIF) in composite health measurement scale: Recommendations for characterizing DIF with meaningful consequences within the Rasch model framework. *PLoS ONE*, *14*(4), e0215073.
- 22. Alroughani, R., & Boyko, A. (2018). Pediatric multiple sclerosis: A review. *BMC Neurology*, *18*(1), 27.
- Buscarinu, M. C., Reniè, R., Morena, E., Romano, C., Bellucci, G., Marrone, A., et al. (2022). Late-onset MS: Disease course and safety-efficacy of DMTS. *Frontiers in Neurology*, 13, 829331.
- Ministry of Housing CLG. (2011). English indices of deprivation 2010 UK: UK Government. https://www.gov.uk/government/stati stics/english-indices-of-deprivation-2010
- Seng, J., Kwan, Y., Fong, W., Phang, J., Lui, N., Thumboo, J., et al. (2020). Validity and reliability of EQ-5D-5L among patients with axial spondyloarthritis in Singapore. *European Journal of Rheumatology*, 7(2), 71–78.
- Bowditch, M. G., Sanderson, P., & Livesey, J. P. (1996). The significance of an absent ankle reflex. *J Bone Jt Surg (Br).*, 78(2), 276–279.
- Vrancken, A. F., Kalmijn, S., Brugman, F., Rinkel, G. J., & Notermans, N. C. (2006). The meaning of distal sensory loss and absent ankle reflexes in relation to age: A meta-analysis. *Journal of Neurology*, 253(5), 578–589.

- Lynch, S., Baker, S., Nashatizadeh, M., Thuringer, A., Thelen, J., & Bruce, J. (2021). Disability measurement in multiple sclerosis patients 55 years and older: What is the Expanded Disability Status Scale really telling clinicians? *Multiple Sclerosis and Related Disorders*, 49, 102724.
- Kavaliunas, A., DanylaitėKarrenbauer, V., Binzer, S., & Hillert, J. (2021). Systematic review of the socioeconomic consequences in patients with multiple sclerosis with different levels of disability and cognitive function. *Frontiers in Neurology*, 12, 737211.
- Merlo, D., Stankovich, J., Bai, C., Kalincik, T., Zhu, C., Gresle, M., et al. (2021). Association between cognitive trajectories and disability progression in patients with relapsing-remitting multiple sclerosis. *Neurology*, *97*(20), e2020–e2031.
- Signori, A., Izquierdo, G., Lugaresi, A., Hupperts, R., Grand'Maison, F., Sola, P., et al. (2018). Long-term disability trajectories in primary progressive MS patients: A latent class growth analysis. *Multiple Sclerosis Journal*, 24(5), 642–652.
- 32. Signori, A., Lorscheider, J., Vukusic, S., Trojano, M., Iaffaldano, P., Hillert, J., et al. (2023). Heterogeneity on long-term disability trajectories in patients with secondary progressive MS: A latent class analysis from Big MS Data network. *Journal of Neurology, Neurosurgery & Psychiatry*, 94(1), 23–30.
- Bosma, A. R., Murley, C., Aspling, J., Hillert, J., et al. (2022). Trajectories of sickness absence and disability pension days among people with multiple sclerosis by type of occupation. *Multiple Sclerosis Journal*, 28(9), 1402–1413.
- Fisher, W. P., Jr., & Stenner, A. J. (2016). Theory-based metrological traceability in education: A reading measurement network. *Measurement: Journal of the International Measurement Confederation*, 92, 489–496.
- 35. World Health Organization. (2001). *International Classification of Functioning, Disability and Health: ICF*. World Health Organization.
- The British Standards Institution (BSI). (2012). Health care services—Quality management systems. Requirements based on EN ISO 9001:2008. UK: The British Standards Institution (BSI).
- International Standard Organisation. (2012). Health informatics— Capacity-based eHealth architecture roadmap. Part 1: Overview of national eHealth initiatives PD ISO/TR 14639–1:2012. 21. 1 ed. UK: International Standard Organisation
- Tremlett, H., Munger, K. L., & Makhani, N. (2021). The multiple sclerosis prodrome: Evidence to action. *Frontiers in Neurology*, *12*, 761408.
- Magyari, M., & Koch-Henriksen, N. (2022). Quantitative effect of sex on disease activity and disability accumulation in multiple sclerosis. *Journal of Neurology, Neurosurgery and Psychiatry*, 93(7), 716–722.
- Prodinger, B., Coenen, M., Hammond, A., Küçükdeveci, A. A., & Tennant, A. (2022). Scale banking for patient-reported outcome measures that measure functioning in rheumatoid arthritis: A daily activities metric. *Arthritis Care & Research (Hoboken)*, 74(4), 579–587.

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.